Author's response to reviews

Title: Simultaneous bicompartmental bucket-handle meniscal tears with intact anterior cruciate ligament: a case report

Authors:

Marios G Lykissas (mariolyk@yahoo.com)
George I Mataliotakis (george.mataliotakis@gmail.com)
Nikolaos Paschos (paschos_nick@yahoo.com)
Christos Panovrakos (orthopuoi@gmail.com)
Alexandros E Beris (aberis@cc.uoi.gr)
Christos D Papageorgiou (chpapage@cc.uoi.gr)

Version: 4 Date: 25 January 2010

Author's response to reviews:

Ioannina, 25-02-2010

Dear Sir or Madam,

After we read carefully the proofs, a final version of the manuscript entitled: “Simultaneous bicompartmental bucket-handle meniscal tears with intact anterior cruciate ligament: a case report” was attached.

Having in mind the recommendations of the Editorial team we proceed to the following changes:

1. The Independent Editorial Production Team: Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. Currently author Nikolaos Paschos is in the manuscript but not the submission system, whilst Christos Panovrakos is present in the submission system but not the manuscript. Author George Mataliotakis also has a middle initial in the manuscript that is not in the submission system. Please amend each author list so that both are identical.

Answer: The list of authors in the manuscript was corrected in order to match exactly the list of authors in the submission system, both in style and order.

2. The Independent Editorial Production Team: Figures: Figures only need to be uploaded via the submission system. Please remove the copies of each figure from the manuscript. The legend and title should be part of the manuscript file, given after the reference list. The figures are numbered automatically in the order in which they are uploaded.

Answer: The figures were removed from the manuscript and uploaded via the submission system.

3. The Independent Editorial Production Team: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make
any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Answer: No further corrections are needed.

Sincerely Yours,

Marios G. Lykissas, MD, (corresponding author)
Department of Orthopaedic Surgery School of Medicine,
University of Ioannina, Ioannina, 45110, Greece.
Phone: ++30 26510 07472, Fax: ++30 26510 34816
E-mail: mariolyk@yahoo.com